Author's response to reviews

Title: Modelling the landscape of palliative care for people with dementia: a European mixed methods study

Authors:

Steve Iliffe (s.iliffe@ucl.ac.uk)
Nathan Davies (nathan.davies.10@ucl.ac.uk)
Myrra Vernooij-Dassen (M.Vernooij-Dassen@iq.umcn.nl)
Jasper van Riet Paap (J.vanRietPaap@iq.umcn.nl)
Ragni Sommerbakk (ragni.sommerbakk@ntnu.no)
Elena Mariani (elena.mariani6@unibo.it)
Birgit Jaspers (Birgit.Jaspers@malteser.org)
Lukas Radbruch (Lukas.Radbruch@malteser.org)
Jill Manthorpe (jill.manthorpe@kcl.ac.uk)
Laura Maio (laura.maio@dementiauk.org)
Dagny Haugen (dagny.haugen@ntnu.no)
Yvonne Engels (Y.Engels@anes.umcn.nl)

Version: 4 Date: 30 July 2013

Author's response to reviews:

In response to the Editorial comment: "I think that the last sentence of the Results section should be removed" we have deleted the sentence in the Results section: 'These will be compared with the existing literature in the Discussion'.